Author’s response to reviews

Title: Differential gene expression in the murine gastric fundus lacking interstitial cells of Cajal

Authors:

Yataro Daigo (yatarod@hotmail.com)
Dr Ichiro Takayama (ichirot@is.icc.u-tokai.ac.jp)
Prof Bruce AJ Ponder (bajp@mole.bio.cam.ac.uk)
Dr Carlos Caldas (cc234@cam.ac.uk)
Sean M Ward (sean@physio.unr.edu)
Kenton M Sanders (kent@physio.unr.edu)
Masayuki A Fujino (mfujino@res.yamanashi-med.ac.jp)

Version: 3 Date: 15 May 2003

PDF covering letter
May 15th, 2003

Dr Enitan Sawyerr
Editorial Assistant
BMC Journals, BioMed Central,
Middlesex House,
34-42 Cleveland Street,
London W1T 4LB,
UK.

RE: BMC Gastroenterology
Dear Dr Enitan Sawyerr

Thank you very much for your decision to accept for publishes of our paper entitled, “Differential gene expression in the murine gastric fundus lacking interstitial cells of Cajal” by Daigo et al. within the BMC Gastroenterology.

Please find the final version of our manuscript. In the version we changed the color of characters all in black. We also attached our tables at the end of the manuscript. As noted in the previous version, we added two authors, Bruce AJ Ponder and Carlos Caldas since they substantially participated in our study: Their contribution is clearly described in the manuscript.

Please do not hesitate to contact us if you have any questions. Thank you very much again for your consideration for our manuscript for publication in BMC Gastroenterology.

Regards,

Ichiro Takayama, MD, PhD.
Dept Gastroenterol, Tokai Univ Sch Med,
Isehara 259-1193, JAPAN.
E-mail: ichirot@is.icc.u-tokai.ac.jp